

Open access • Journal Article • DOI:10.1353/JHR.2008.0005

Out of the Wallet and into the Purse Using Micro Data to Test Income Pooling

— Source link < □</p>

Jennifer Ward-Batts

Institutions: Claremont McKenna College

Published on: 31 Mar 2008 - Journal of Human Resources (University of Wisconsin Press)

Topics: Family income, Distribution (economics), Pooling and Child benefit

Related papers:

· Do husbands and wives pool their resources?: evidence from the United Kingdom child benefit

• Intra-household resource allocation: an inferential approach

· Collective Labor Supply and Welfare

• Tests of income pooling in household decisions

• Nash-bargained household decisions: toward a generalization of the theory of demand







Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Ward-Batts, Jennifer

Working Paper

Out of the Wallet and into the Purse: Using Micro Data to Test Income Pooling

Claremont Colleges Working Papers, No. 2003-10

Provided in Cooperation with:

Department of Economics, Claremont McKenna College

Suggested Citation: Ward-Batts, Jennifer (2003): Out of the Wallet and into the Purse: Using Micro Data to Test Income Pooling, Claremont Colleges Working Papers, No. 2003-10, Claremont McKenna College, Department of Economics, Claremont, CA

This Version is available at: http://hdl.handle.net/10419/23385

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Out of the Wallet and into the Purse: Using Micro Data to Test Income Pooling

Jennifer Ward-Batts*

May 2003

Abstract:

Unitary models, assuming a single objective function and unified budget constraint, are traditionally used to model household behavior. Most empirical tests of unitary models rely on endogenous regressors. This paper uses an exogenous change in the intrahousehold distribution of income, provided by a change in U.K. Family Allowance policy. Expenditure shares are estimated for a wide range of goods. Shifts in expenditure shares for assignable goods, such as men's clothing, children's clothing, and men's tobacco, suggest that children benefited at the expense of men when this policy change shifted income within households from men to women.

JEL codes:

D12: Household Behavior and Family Economics – Consumer Economics: Empirical Analysis

D19: Household Behavior and Family Economics – Other

D79: Analysis of Collective Decision-Making – Other

138: Welfare and Poverty – Provision and Effects of Welfare Programs

J18: Demographic Economics – Public Policy

Keywords: income pooling, intrahousehold allocation, child benefit, collective model, unitary model, family policy, household demand

^{*}Claremont McKenna College, Dept. of Economics, 500 E 9th St, Claremont, CA 91711, email: jennifer.ward-batts@mckenna.edu. I am grateful to Shelly Lundberg, Richard Startz, Robert Pollak, Terry Wales, Yoram Barzel, Elaina Rose, Duncan Thomas, and numerous seminar participants for useful discussions and comments. All opinions and errors are my own. I am grateful for support from an Alfed P. Sloan Doctoral Dissertation Fellowship while working on the earliest draft of this work. Material from the Family Expenditure Survey is Crown Copyright, has been made available by the Central Statistical Office through the ESRC Data Archive, and has been used by permission. Neither the CSO nor the ESRC Data Archive bear any responsibility for the analysis or interpretation of the data herein.

I. Introduction

In the past two decades, there has been much discussion of, and many attempts to test, the "pooling hypothesis" implied by unitary models of household decision-making against the set of alternative collective models.¹ Unitary models assert that a household behaves like an individual, maximizing a single objective function subject to a unified budget constraint. One implication of these models is that household members pool their income, and that who controls what proportion of that income does not affect household demands.

Empirical tests of income pooling are typically plagued by potential biases due to the endogeneity of measures of control over income in the household. In cross section data, the magnitude and distribution of income in the household may be correlated with the preferences of and prices faced by family members. Clearly what is needed for a test of pooling is an experiment generating variation in intra-household income control. In the late 1970s, the U.K. changed the form of its universal child benefit scheme, essentially shifting receipt of transfer income from fathers to mothers in two-parent families. This "natural experiment" provides an exogenous source of variation in the control of resources within the family. Using aggregated data, grouped by household composition and income, Lundberg, Pollak, and Wales (1997) (LPW) find that ratios of children's to men's and of women's to men's clothing expenditures in households with children in the U.K increased after this policy change took effect. This study did not control for changes in relative prices of the three goods, and was limited in the goods examined.

-

¹ Lundberg and Pollak (1996) provide a comprehensive discussion of both the theoretical and empirical literature.

In this paper, I use household-level data to test for changes in expenditure patterns for a broader range of goods, using a single-difference model in a time series of cross sections.² I estimate the effect of the policy change on the share of the budget allocated to each of a comprehensive set of eleven broadly defined goods categories. The hypothesis that expenditure shares for all these goods were unaffected by the shift in income control is strongly rejected. While any significant change in expenditure patterns attributable to this policy change warrants a rejection of income pooling, testing the pooling hypothesis using goods that are assignable to particular household members is comparatively more appealing.

I estimate budget shares for men's, women's, and children's clothing, controlling for prices of each of these goods³. Results are consistent with those of LPW. I also estimate budget shares for several narrowly defined goods that may be of greater interest either to men or to women, or of particular relevance to child welfare. I find shifts in expenditures on some quasi-assignable goods that are generally in the direction a collective model would predict, given the direction of change in control over household income. For instance, men's clothing and a men's tobacco category - consisting of cigars, pipe tobacco, and snuff products - decline, while children's clothing, toys, and pocket money, and restaurant and take-away meals increase as a share of total expenditure.

The paper is organized as follows: Section two provides some theoretical and empirical background motivating this research, including a discussion of the change in the Family

_

² It seems appealing to use married couples with no children in the household as a control group, since they were unaffected by this policy change. I explored this possibility for all the goods examined here. I find, for the majority of these goods, that I cannot reject the null hypothesis that the time trends in the four years before the policy change differ between these two groups, and thus conclude that couples with no children are not a good control group. That group contains couples who are childless and will remain so, couples who have not yet had children, and those who have already raised their children. In these data, it is impossible to distinguish these from one another. It is plausible that any of these three groups may behave differently than couples with children, say, with respect to women's labor force participation for example, and thus have different time trends in expenditures.

Allowance policy. Section three discusses the data; and section four, the empirical models. Section five presents results and their implications; and section six concludes.

II. Background

Traditional economic theory avoids the difficult issue of how decisions are made in households by assuming each household maximizes a single objective function subject to a unified budget constraint. Samuelson (1956) suggests that this objective function might be arrived at through consensus among family members. Becker (1981) proposes it may represent the preferences of an altruist (dictator) in the household. The restrictions imposed by the unitary models of Samuelson and Becker are strong ones. The household objective function is assumed to have the properties of a utility function, and to be invariant to who controls resources in the household. One restriction implied by such models is that household members pool their resources. This pooled income is then used to maximize the household objective function. This implies that only total household income, not its distribution, affects household demands.

This has important practical implications for policies targeted at improving the welfare of particular household members, such as women or children. Income pooling results in transfers to targeted household members being neutralized by the household allocation mechanism. The welfare of the targeted member may be improved, but no more or less than if the transfer had been given to another household member. Cash transfers, as well as transfers in kind, may be subject to this reallocation.

The concept of individual rational choice, which lies at the heart of microeconomic theory, compels us to consider the possibility that each person may have distinct preferences, and

³ LPW used only an aggregate clothing category price index, while this study utilizes separate price indices for each

that there may therefore be competing interests of individuals within a household. Social choice theory has demonstrated that aggregating preferences is no simple matter, and that the aggregate function arrived at may be sensitive to the mechanism of aggregation.

Relatively recent theoretical work allows for the possibility that each adult in the household has distinct preferences, and analyzes how their competing interests are reconciled.⁴ Models which do not impose pooling have been introduced, including general collective models (Chiappori 1988, 1992), and both cooperative and non-cooperative bargaining models due to Manser and Brown (1980), McElroy and Horney (1981), Bergstrom (1996), and Lundberg and Pollak (1993, 1994), among others. These models allow income accruing to different family members to affect household demands differently. For instance, in Nash bargained solutions, each person's (potential) income affects his or her reservation level of utility, or threat point, and therefore his or her equilibrium level of utility in the household. Thus, these models allow for more effective targeted transfer policies. In cases where the difference in predictions of and unitary models is important, and if unitary models make poor predictions, then collective household models must be considered.

Endogeneity of the distribution of income within the household creates obstacles to testing the pooling hypothesis. An obvious example is earned income. Its distribution and magnitude depend on hours worked by each spouse, which are determined jointly with household expenditures. Unearned income is also likely to be endogenous with respect to household behavior. For instance, income from assets may be correlated with past labor supply, and thus with current labor supply as well. Transfers, both public and private, are typically conditioned on some set of criteria over which household members exercise some control.

of the three sub-categories.

Sources of unexpected unearned income that are not subject to these concerns tend to be sporadic and insignificant. Previous attempts to test the pooling hypothesis, including Thomas (1990), Schultz (1990), Phipps and Burton (1998), and Bourguinon, Browning, Chiappori, and Lechene (1993) have used some of these problematic sources of income. These tests have generally rejected pooling, but possible endogeneity biases call those results into question.

Another hurdle in testing for income pooling is that consumption data are typically collected at the household, not the individual, level. Though individual expenditures are collected by some surveys, the purchaser is not necessarily the consumer. We cannot assign consumption of most goods to any one person, and the possibility of interdependent preferences makes it even more problematic to assign utility from consumption. However, complete assignability is not required. Examining goods which are of greater interest to some household members than to others, such as the three clothing categories examined by LPW, provides an intuitive means of testing the pooling hypothesis, and whether any shifts in expenditure patterns are consistent with what we would expect in the context of collective models.

In the United Kingdom in the late 1970s, the child benefit scheme was altered in such a way as to shift income from fathers to mothers in two-parent households. Prior to 1977, the universal child benefit scheme in the U.K. consisted of a small taxable Family Allowance payment to the mother, and a more significant Child Tax Allowance, which reduced the amount of taxes withheld from earned income. The latter generally would have increased the father's take home pay. This two-part program was phased out over the period April 1977 to April 1979 and replaced by the Child Benefit, a non-taxable cash payment to the mother. While the average amount of the total benefit stayed roughly constant, this policy change shifted apparent control

⁴ Carolyn Moehling (1997) extends this to include children as household decision-makers.

of a portion of family income from fathers to mothers. The amount of income involved is a significant fraction of the average family's budget. For example, for a family with two children in April 1980, the Child Benefit was approximately 445 British pounds per year⁵ (approximately 180 pounds in 1974 currency), or about 8% of male manual earnings in the UK.⁶ The real value of that benefit is 7.5% of average total expenditure among families with one to three children in the data used for this study. In April 1974, the Family Allowance paid to the mother in a two-child family was 47 pounds. Since Family Allowance was taxable and Child Benefit was not, the difference of 134 pounds can be viewed as an upper limit of the amount of income shifted to the wife.

Based on this policy shift, which is clearly exogenous with respect to individual household expenditure decisions, LPW present empirical evidence against the pooling hypothesis using grouped data on household expenditures. Using expenditure data from before and after the policy change, they "find strong evidence that a shift towards relatively greater expenditures on women's goods and children's goods coincided with this income redistribution." (p.1) Limitations of the aggregated data dictated some shortcomings, such as a concentration on clothing, and limited demographic controls. Each cell in this data consists of the mean of expenditures on a category of goods for all families in the sample which fall into a particular income—family-size group. The small number of cells in the data make it difficult to distinguish effects of the policy change from time trends. Another potential confounder is the 1979 change in the value-added tax (VAT) rate. This rate rose from 8% to 15%, and at the same time, clothing and footwear for young children gained zero-rated status. This change, occurring

_

⁵ Social Security Statistics (1991), Table G1.01, p. 253.

⁶ U.K. House of Commons Hansard, 14 January 1980, pp. 641-2.

⁷ See Lundberg, Pollak, and Wales (1995) for a more detailed discussion of the policy change.

⁸ LPW did examine alcohol and tobacco expenditures but did not find significant changes.

around the time of the change in the Child Benefit, likely affected the relative price of adult and children's clothing. LPW use a price index for the aggregate category, clothing and footwear. Thus, the decline they find in the ratio of children's clothing expenditures to men's clothing expenditures may be attributable in part to changes in relative prices for which they are unable to control.

I use separate price indices for the three clothing categories in estimates presented here. Since a change in the relative price of children's clothing is of primary concern in connection with the VAT change, this addresses whether mothers appear to allocate more resources to children than do fathers, as has been suggested by a number of other studies examining intrahousehold allocation issues.

Using household-level data from the Family Expenditure Survey, I estimate budget shares for some narrow goods categories which may be to some extent assignable to certain members of the household, including men's, women's, and children's clothing and footwear, and some other narrowly defined goods, such as men's tobacco products. Since the policy change gives wives control over a greater proportion of household income, we would expect to see a shift in consumption toward goods which are of relatively greater interest to wives. Such shifts would warrant a rejection of pooling, and thus provide evidence against the unitary model. The nature of these shifts also have important implications for the effectiveness of policies intended to improve welfare for targeted household members.

III. Data and Sample Construction

The Family Expenditure Survey is conducted annually in the U.K. Ten thousand households are randomly selected each year, approximately seventy percent of which complete the survey. One or more face-to-face interviews are conducted with one or more members of each household. In addition to these interviews, each "spender" in the household completes a personal spending diary for a period of two weeks. From the interviews and the diaries, detailed information on household expenditures, income, and household demographics is compiled. Some expenditure items, such as food and clothing, are covered only in the diaries. For other items, such as fuel and housing expenditures, detailed interview questions supplement the information from the diaries. Expenditures are reported as weekly values.

The survey is spread over the entire the year, making it possible to control for seasonal effects on expenditures. The data include the region in which the household is located: Northern Ireland, Scotland, Wales, or one of nine regions in England. The data are collected by the Office of Population Censuses and Surveys Social Survey Division for the Department of Employment. The data archive is maintained by the Economic and Social Research Council Data Archive at the University of Essex.¹⁰

The primary purpose of the survey is to provide the weights for construction of the Retail Price Index. However, the data are intended to also be useful for research purposes. They include detailed income information by source and by household member. The age and sex of each household member are reported, as well as the number of persons by sex in each of several age categories. I use these categories for children (aged under 2 years, aged 2 but less than 5,

-

⁹ A spender is defined as anyone at least age 15 prior to 1973, or at least age 16 beginning in 1973.

¹⁰ Material from the Family Expenditure Survey is Crown Copyright, has been made available by the Central Statistical Office through the ESRC Data Archive, and has been used by permission. Neither the CSO nor the ESRC Data Archive bear any responsibility for the analysis or interpretation of the data reported here.

aged 5 but less than 16, and aged 16 but less than 18) as a general guide for constructing dummy variables to control for household composition. I create eight categories defined by the number of children in the household and their ages.

Approximately 7,000 households complete the Family Expenditure Survey in any given year. For this analysis, the sample is limited to households with one man, one woman who is recorded as the wife of the household head and less than age 60, and one to three children¹¹. This results in a total sample size of 15,810 households for 8 years of data. I use data from 1973 to 1983 but drop the intermediate years, 1977-1979, during which the Child Benefit policy change was phased in. I create three binary variables for the post-policy-change period, one for each family size, allowing for the policy to affect families of different sizes in a general way.

Price indices from the Abstract of National Statistics are used. These indices are available quarterly for 11 broadly defined goods categories, as well as for all goods combined (the Retail Price Index). In estimates of narrowly defined clothing expenditures, I use detailed price indices obtained from the Office for National Statistics in the U.K. These indices are available monthly for a number of narrowly defined goods, including men's clothing, women's clothing, and children's clothing. 12 The two sets of price indices come from the same source, but the latter is aggregated to a lesser degree. Neither set is available separately by region.

Adults are defined as all persons age 18 and over, and married persons irrespective of age.
 The children's clothing price index is unavailable prior to 1974.

III. Empirical Models

The basic empirical model is an adaptation of the Almost Ideal Demand System (AIDS) of Deaton and Muellbauer (1980a,b). The budget share for each good can be estimated equation-by-equation using:

$$w_{ig} = \alpha_g + \gamma \log p_g + \beta \log \left(\frac{C_i}{I_i}\right) + f(D_i, H_i, Q_i, t_i) + u_{ig}$$
(1)

where w_{ig} is the budget share for good category g for household i; p_g is the relative price for good category g, computed by dividing the national price index for that category by the Retail Price Index. ¹³ C_i is total household consumption or expenditure. I_i is the household-specific price index for all goods, approximated by $\log I_i = \sum_g w_{ig} \log p_g$. ¹⁴ D is a set of dummy variables representing the policy change, and H is a vector of household demographic variables. Q is the quarter of the year in which the household was surveyed, t measures a time trend, and u_{ig} is an error term with standard properties. This model can be estimated equation-by-equation using ordinary least squares regression.

Given expenditure diary data for a two-week period, many households will record zero expenditures on some goods. The correct empirical model to apply depends on the mechanism generating those zeroes. If a household does not consume a particular good, as may be the case with alcohol or tobacco, then the tobit model can be applied. However, if a good is consumed by the household but not purchased during the period, such as in the case of clothing, then the tobit

11

¹³ While inclusion of all prices is preferred, multicolinearity results if all prices are included due to the short time series and absence of cross-sectional (regional) price variation. Narrowly defined clothing estimates do include all three narrow clothing prices.

¹⁴ According to Deaton and Muellbauer (1980b) this approximation was used by J. R. N. Stone. No particular work is cited. I use broad category expenditures and corresponding price indices to calculate this index.

model is not the correct specification. In this case, OLS is consistent under fairly weak conditions.¹⁵

I use OLS to estimate the model in (1) for all goods. I also present tobit estimates of the model for goods which a substantial fraction of households may choose to not consume, such as alcohol and tobacco. If both non-consumption and purchase frequency play a role, then the truth is probably somewhere between the results of the two models.

Instrumental Variables Estimation to Correct for Measurement Error

As above, let us assume that expenditure and consumption are equal on average for a sufficiently long time period, in essence that $E(y_{ig}) = E(y_{ig}^*)$ for household i, good g. If we observe expenditure by a household for a random two-week period, we can write the relationship of observed expenditure to unobserved consumption as $y^*_{ig} = y_{ig} + e_{ig}$, where e_{ig} can be thought of as a mean-zero measurement error. This implies that total expenditure will be higher in periods when clothing is purchased than in periods when it is not. Normally, measurement error in the dependent variable is not a concern, but e_{ig} will also enter in total consumption, which is a regressor. For simplicity, suppose only consumption of good 1 is measured with error, so that total consumption equals total expenditure plus e_{if} :

Total Consumption
$$\equiv C_i \equiv \sum_g y_{ig}^* = \sum_g y_{ig} + e_{i1}$$
 where Total Expenditure $\equiv E_i \equiv \sum_g y_{ig}$

We can substitute into equation (1) and rearrange terms to obtain:

-

¹⁵ A purchase infrequency model (see Cragg 1971or Blundell and Meghir 1987) might be used, but there are problems with applying it here. First, it is difficult to find convincing exclusion restrictions – i.e., measured variables that affect consumption but not frequency of purchase, or vice versa. Second, for purchases made on credit, only payments toward the purchase, not the total purchase amount, get recorded in the data. Third, zeroes may be observed for some goods in the data, such as alcohol and tobacco, for both non-consumption and purchase infrequency reasons. A double-hurdle model allows for both, but the two portions of the model cannot be convincingly identified using these data.

$$\frac{y_{i1}}{E_i} = \frac{E_i + e_{i1}}{E_i} \left[\alpha_1 + \gamma \log p_1 + \beta \log \left(\frac{E_i + e_{i1}}{P_i} \right) + f(D_i, H_i, Q_i, t_i) \right] + v_{i1}$$
(2)

where
$$v_{i1} = u_{i1} + \frac{e_{i1}(u_{i1} - 1)}{E_i}$$
 (3)

This gives us the measured budget share on the left-hand side. Note that the regressor, total consumption, is correlated with the error term through the presence of e_{i1} in both terms. We can instrument for total expenditure where it is used as a regressor intended to measure total consumption, arguably getting closer to a true measure of total consumption and doing away with some of the measurement error. However, some correlation between the regressor and error term will remain because total expenditure appears in the denominator of the latter. Further, estimated coefficients on regressors will be scaled by total consumption over total expenditure. However, as long as these are equal on average, this ratio is unity.

Because the specification of the AIDS model calls for the consumption share and not the level of expenditure to be estimated, elimination of measurement error bias in estimates is not assured by instrumenting for total expenditure. Note that if we estimate consumption levels instead, our equation would be the following:

$$y_{i1} = \alpha_{i1} + \gamma \log p_1 + \beta \log \left(\frac{E_i + e_{i1}}{I_i} \right) + f(D_i, H_i, Q_i, t_i) + \varepsilon_{i1} \quad \text{where} \quad \varepsilon_{i1} = u_{i1} - e_{i1}$$

$$\tag{4}$$

In this case, instrumenting for total expenditure is sufficient to eliminate the bias, given that instruments used are appropriate.

Keen (1986) uses "normal income" (based on usual rather than current earnings) to instrument for total expenditure in estimating Engel curves using a single year of FES data, and shows that this produces consistent estimates. If macroeconomic conditions are constant, then consumption should be approximately proportional to normal income. However, changes in

macroeconomic conditions may affect saving rates and therefore the proportion of normal income consumed in a given period.

I use normal income and regional quarterly unemployment rates as instruments for total expenditure. Because normal income is zero in some cases, a log specification, to match that for total expenditure, is not feasible. I find that a fifth-order polynomial in normal income and a third-order polynomial in unemployment rate provides the best fit. These variables are used as instruments in two-stage least-squares estimates of both the budget share and the level of expenditure for each of the three clothing categories

If clothing is separable from other goods, and if clothing is the only good for which consumption and expenditure during the survey period may differ, then standard OLS and tobit models will suffice for other goods. These are strong assumptions. Clearly, other goods also suffer from purchase infrequency. These may include durables, haircuts, toys, and books to name a few. Therefore instrumental variables methods will also be used in estimating expenditure shares and levels for eleven broadly defined goods categories and nine additional narrowly defined goods. . In cases where a tobit model should be used, I estimate total expenditure using the instruments discussed above as regressors in an OLS model and enter the estimated total expenditure in the tobit model. I'll refer to this as a two-stage tobit model. ¹⁶ OLS estimates of total expenditure using the instruments as regressors (the first stage) is shown in Appendix A. All regressors are significant at the one percent level or better, with the exception of the quadratic normal income term. The adjusted R squared for the model is 0.36.

Recall that in order for the measurement error to be mean zero, we require a sufficiently long time period so that average consumption equals average expenditure. The length of this

¹⁶ Standard errors have not been corrected in two-stage tobit estimates. This is unlikely to affect inference for the

time period may differ for different goods. In particular, I will assume that three months is not a sufficiently long period for clothing, and so will not control for quarter of the year in clothing estimates.¹⁷ Though I do control for quarter in estimates of other goods, I check whether effects of the policy change are affected by inclusion versus exclusion of these variables for goods for which a quarter is arguably too short to assume expenditure equals consumption, such as durables, books, and toys.

An additional method will also be used which allows estimation of the budget share while minimizing concerns about appearance of total expenditure in the denominator of the error term. I subtract clothing and durables expenditures from total expenditure and use the result to proxy for total expenditure, both on the right-hand side and in the share calculation. Standard OLS and tobit models are then used to estimate budget shares. These results are compared to IV results as a check of robustness.

IV. Results

I estimate budget shares and expenditure levels for eleven broadly defined goods categories and twelve narrowly defined goods, including three assignable clothing categories. Table 1 presents summary statistics, including means of total expenditure and income measures, and expenditure shares and levels for the goods of interest. The number of households reporting positive expenditures in each good category is also given. The distribution of families across the three family-size categories is shown. Almost half of families in the sample have two children.

policy change variables which are the focus of this paper.

¹⁷ Clothing consumption need not be greater in colder months. We may wear more clothes at once in colder months, but may use as many clothing services in warm months due to summer activities and more frequent washing, especially in the case of children's clothes.

Thus, the omitted household composition category in estimates is the most common child-age category among two-child families.

Table 2 presents results for instrumental variable estimates of the eleven broad goods categories. Panels (a) and (c) show budget share estimates while panels (b) and (d) show expenditure level estimates. Because more than five percent of households report zero expenditures on alcohol, clothing, food out and tobacco, both two-stage least squares and two-stage tobit estimates are reported for these two categories (panels c and d). TSLS estimates are reported for the remaining seven categories (panels a and b): durables, food for home preparation (food in), fuel and power (not including fuel for automobiles), housing, services, transportation (including petrol), and miscellaneous goods. The three policy shift variables (for one-, two-, and three-child families) had a positive and significant effect on expenditure shares and levels for clothing, food out, and miscellaneous goods. It had negative and significant effects on expenditure shares and levels for housing. There is also some evidence of decline in alcohol expenditures, particularly in three-child families, although this result is not significant in tobit models, which are theoretically more appealing given that there may be non-consuming households.

Estimates for broadly defined goods are done equation by equation, and no constraint on adding up of shares is imposed.¹⁸ F-tests and likelihood ratio tests were used for each of the broad goods to test the joint hypothesis that the three policy shift dummies had no effect. These tests indicate that the policy shift did significantly affect budget shares for several of the categories, increasing clothing, food out, and miscellaneous goods expenditures and decreasing expenditures on housing and "food in." The durables estimates were also done excluding

1

¹⁸ The sum of TSLS policy change coefficients for budget shares for a given size family is approximately zero.

controls for quarter of the year and the results on policy variables, which are not statistically different from zero, were virtually unchanged. Table 2 also reports a chi-square statistic based on a seemingly unrelated regressions system of equations (estimated with instruments using three-stage least squares) testing the joint hypothesis that effects of the three policy variables in all eleven equations are zero. This hypothesis is strongly rejected, indicating that the policy shift did significantly change overall expenditure patterns.

Phipps and Burton (1998) and Hoddinott and Haddad (1991) found negative effects of the wife's relative income on tobacco and alcohol. I find marginal negative effects for alcohol. Though I find only slight evidence of negative effects for all tobacco, I do find negative effects for a men's tobacco category, which excludes cigarettes. Phipps and Burton also found positive effects of the wife's income on restaurant meals. This could be interpreted as a price effect since the study uses earned income as the key explanatory variable. The exogeneity of the policy change in the present analysis rules out such an interpretation here of the strong positive effect on "food out," which includes restaurant meals and take-away food.

An alternate method in which total expenditure less clothing and durables expenditures is used as a proxy measure of total consumption in estimating budget shares yields somewhat similar results (shown in Table B1 of Appendix B) to those on broadly-defined goods using IV methods. In general, signs of coefficients on the policy variables are the same, but statistical significance in most cases is stronger. Some differences from the IV models should be highlighted: Here, the negative coefficients for tobacco in one- and two-child families is significant in the OLS model, but not the tobit model. To the extent that purchase infrequency is a more important generator of zeroes than non-consumption, the OLS estimates are preferred to the tobit. Alcohol does decline significantly here for one- and three-child families in the tobit

model, and the negative coefficients in the "food in" equation are significant for all family sizes. Two stark differences emerge – effects on durables here are positive and significant and effects on fuel and power negative and significant. This differs markedly from the IV models, and involves switching signs of coefficients, although they were far from statistically different from zero in the earlier models. Again, a seemingly unrelated regressions system of equations is used to test the joint hypothesis that policy variables in all equations have zero effect, and again the hypothesis is strongly rejected (p<.001).

Table 3 reports results from expenditure share equations on nine narrowly defined goods. All of these goods had more than five percent zero expenditures reported, so both OLS and tobit results are presented. For goods which are arguably consumed by most or all households in the sample but which are purchased infrequently, least squares estimates are theoretically more appealing, as discussed earlier. However, it may be that, for some of these goods, there is both non-consumption and infrequent purchase (e.g., pets expenditures). A model which distinguishes these empirically cannot be convincingly identified with this data. I therefore present both two-stage least squares and two-stage tobit estimates.

Cigarettes expenditures do not appear to have been affected by the policy change, but other tobacco expenditures did decline, significantly for two-child families, which comprise almost half of the sample. The joint test of all three policy variables having no effect on other tobacco is rejected in the levels estimates. This category includes cigars and pipe tobacco, almost exclusively consumed by men, whereas cigarettes are a gender-neutral good, widely consumed by both men and women in the UK.

Domestic services, which might be viewed as a substitute for the wife's home production, appear to have increased for one-child families per the tobit results, but declined for two-child

families per the least squares model. There are most certainly non-consumers, but purchase frequency may also play a role. The significant joint test statistics are puzzling and not very informative given that coefficient signs differ for different family sizes. Cosmetics expenditures decline significantly. While these goods are directly consumed mostly by women, one might argue that the husband derives utility from their use, or that their use by the wife may elevate her relative bargaining power by increasing the value of her outside options relative to those of her husband. Hairdressing includes trips to the barber or stylist for all family members, and does not appear to be affected by the policy change.

In panels (c) and (d), we see a positive effect on books, but this is only significant in tobit models. Children's books cannot be distinguished from adult books in the data. The books category also includes some other reading materials, such as periodicals. While it is unlikely there are many households in this sample who don't consume in this category, consumption may be via borrowed library materials, so that expenditure and consumption do not relate to one another in the way I have specified above. There is also most probably an important frequency of purchase element here.

Pets expenditures appear unaffected, while toys are increasing for one- and two-child families. There are likely few non-consumers of toys in this sample, but purchase frequency is likely to be important. Children's pocket money is increasing, significantly for one- and (in one model) two-child families. Results from proxied OLS models for these nine narrowly-defined goods are quite similar to those in instrumented estimates (see Table B2 of Appendix B).

Table 4 reports two-stage least squares shares and levels estimates, and proxied OLS estimates for the three narrow clothing categories. All three models show strong and statistically significant positive effects of the policy change on children's clothing. Policy change

coefficients are consistently negative for men's clothing in all three models, and the effect is statistically significant for three-child families in both instrumented models. The coefficients are consistently positive for women's clothing across all three models, and are statistically significant in two- and three-child families in the proxied OLS model. A test of the joint hypothesis for all three policy variables for women's clothing shows a statistically significant effect in both the instrumented share estimates and the proxied share estimates.

Results found here on men's, women's, and children's clothing are consistent with those found by Lundberg, Pollak, and Wales (1997). They found that the ratio of children's clothing expenditures to men's clothing expenditures increased when the child benefit policy changed. They also found an increase in the ratio of expenditures on women's clothing to that on men's clothing. As discussed earlier, their results on children's relative to men's clothing may be attributable in part to the 1979 change in VAT rates on children's relative to adult clothing. 19 The decline they find in men's clothing relative to women's clothing is unlikely to have been driven by the change in tax rates, since the tax rate change on both goods was the same, but the tax inclusive price of children's clothing relative to adult clothing may have declined significantly. Thus, their result on children's relative to men's clothing expenditures is of particular concern. The results I have presented here are compelling given that narrow price indices with VAT included were used to control for prices of the three separate clothing categories. These results demonstrate that the increase in the relative share of children's goods they find was not solely due to tax rate changes, but is attributable in part to the child benefit policy change, and to the apparent preference of mothers to allocate more of the family budget to children's goods.

10

¹⁹ VAT rates were 10% in 1973, changing to 8% in 1974, and to 15% in 1979. Young children's clothing and

The magnitudes of changes in budget shares these results suggest are certainly plausible given the amount of income being shifted from men to women. For instance, for the average two-child family, the change in expenditures on the broad category "food out" attributable to the policy change is £26 per year. As a result of the policy change, the same family decreased its annual housing expenditure by £90 and men's tobacco expenditure by £3, and increased children's clothing expenditures by £16 and toys expenditures by £5. These are reasonable magnitudes given that approximately £134 per year was shifted from husband to wife.²⁰

V. Conclusion

I use micro data to examine how budget shares changed in two-parent families with children in the U.K. when a change in Family Allowance policy essentially shifted transfer income from fathers to mothers. Shifts in expenditure patterns due to this change are inconsistent with the unitary model of household decisions. I find significant changes in expenditures on broadly defined goods, as well as on some narrowly defined assignable and quasi-assignable goods. Although the broad goods categories do not represent assignable goods, a discrete change in shares in the before versus after period which is not attributable to other factors indicates that there has been a shift of power over decision making in the household, and that a systematic difference in preferences over allocation of household income exists between husbands and wives. Closer examination of expenditures on narrow goods within these broad categories produces further insights. These shifts indicate that women and children benefited at the expense of men when this new policy took effect.

footwear became zero-rated in 1979. (Prest, 1980)

²⁰ The quoted magnitudes of change are in 1974 currency, and are based on TSLS budget share estimates.

Results on narrow goods generally point to an increase in the budget share allocated to some women's and children's goods, along with a decline in that for some men's goods. Such changes in consumption patterns resulting from a shift in the control of income in the household adds evidence to the growing case against the unitary model — evidence which is particularly convincing given the exogeneity of this policy shift.

Rejection of the unitary model has important implications for the effectiveness of policies aimed at improving welfare of targeted members of households. The unitary model implies neutralization of targeted transfers. Some alternative collective models allow for more effective transfer policy. As Alderman, et. al. put it, it may be "time to shift the burden of proof."

Analyzing the household allocation process within the framework of collective models will allow the development and application of more effective policies in areas of health, education, and welfare.

References

- Alderman, Harold, Chiappori, Pierre-Andre, Haddad, Lawrence, Hoddinott, John and Kanbur, Ravi (1995). 'Unitary versus collective models of the household: is it time to shift the burden of proof?' *World Bank Research Observer* vol. 10(1), pp. 1-19.
- Becker, Gary (1991). *A Treatise on the Family*, enlarged edition. Cambridge, Mass.: Harvard University Press.
- Bergstrom, Theodore (1996). 'Chapter 2: A survey of theories of the family,' in (Mark R. Rosenzweig and Oded Stark, eds.) *Handbook of Population and Family Economics*, vol. 1A, Amsterdam: North-Holland.
- Blundell, Richard and Meghir, Costas (1987). 'Bivariate alternatives to the tobit model,' *Journal of Econometrics*, vol. 34(2), pp. 179-200.
- Bourguignon, Francois, Browning, Martin, Chiappori, Pierre-Andre and Lechene, Valerie (1993). 'Intra household allocation of consumption: a model and some evidence from French data,' *Annales d'Economie et de Statistique*, vol. 29, pp. 137-156.
- Chiappori, Pierre-Andre (1988). 'Rational household labor supply,' *Econometrica*, vol. 56(1), pp. 63-90.
- Chiappori, Pierre-Andre (1992). 'Collective labor supply and welfare,' *Journal of Political Economy*, vol. 100(3), pp. 437-467.
- Cragg, John C. (1971). 'Some statistical models for limited dependent variables with application to the demand for durable goods,' *Econometrica*, vol. 39(5), pp. 829-844.
- Deaton, Angus and Muellbauer, John (1980a). 'An almost ideal demand system,' *American Economic Review*, vol. 70(3), pp. 312-326.
- Deaton, Angus and Muellbauer, John (1980b). *Economics and Consumer Behavior*. New York: Cambridge University Press.
- Deaton, Angus and Irish, Margaret (1982). 'Statistical models for zero expenditures in household budgets,' *Journal of Public Economics*, vol. 23(1), pp. 59-80.
- Hoddinott, John and Haddad, Lawrence (1995). 'Does female income share influence household expenditures: evidence from The Cote d'Ivoire,' *Oxford Bulletin of Economics and Statistics*, 57(1), pp. 77-96.
- Kay, J.A., Keen, M.J. and Morris, C.N. (1984). 'Estimating consumption from expenditure data,' *Journal of Public Economics*, vol. 23(2), pp. 169-181.
- Keen, Michael (1986). 'Zero expenditures and the estimation of Engel curves,' *Journal of Applied Econometrics*, vol. 1(3), 277-286.

- Kemsley, W.F.F. (1969), Family Expenditure Survey Handbook on the Sample, Fieldwork, and Coding Procedures. London: HMSO.
- Kemsley, W.F.F. (1980), Family Expenditure Survey Handbook on the Sample, Fieldwork, and Coding Procedures, revised edition. London: HMSO.
- Lundberg, Shelly J. and Pollak, Robert A. (1993). 'Separate spheres bargaining and the marriage market,' *Journal of Political Economy*, vol. 101(6), pp. 988-1010.
- Lundberg, Shelly J. and Pollak, Robert A (1994). 'Noncooperative bargaining models of marriage,' *American Economic Review*, vol. 84(2), pp. 132-137.
- Lundberg, Shelly J. and Robert A. Pollak (1996). 'Bargaining and distribution in marriage,' *Journal of Economic Perspectives*, vol. 10(4), pp. 139-158.
- Lundberg, Shelly J., Pollak, Robert A. and Wales, Terry J. (1997). 'Do husbands and wives pool their resources? Evidence from the U.K. child benefit,' *Journal of Human Resources*, vol. 32 (3), pp. 463-480.
- Manser, Marilyn and Brown, Murray (1980). 'Marriage and household decision making: a bargaining analysis,' *International Economic Review*, vol. 21(1), pp. 31-44.
- McElroy, Marjorie B. and Horney, Mary Jean (1981). 'Nash bargained household decisions,' *International Economic Review*, vol. 22(2), pp. 333-349.
- Meghir, Costas and Robin, Jean-Marc (1992). 'Frequency of purchases and the estimation of demand systems,' *Journal of Econometrics*, vol. 53(1), pp. 53-85.
- Moehling, Carolyn (1997). 'The impact of children's income on intrahousehold resource allocations,' prepared for the NBER/DAE 1997 Summer Institute.
- Phipps, Shelley and Burton, Peter (1998). 'What's mine is yours? The influence of male and female incomes on patterns of household expenditure,' *Economica*, vol. 65(260), pp. 599-613.
- Pollak, Robert (1994). 'Taking power seriously,' mimeo, University of Washington.
- Prest, A.R. (1980). *Value Added Taxation: The Experience of the United Kingdom*, Washington: American Enterprise Institute for Public Policy Research.
- Robin, Jean-Marc (1993). 'Economic analysis of the short-run fluctuations of households' purchases,' *Journal of Economic Studies*, vol. 60, pp. 923-934.
- Samuelson, Paul A. (1956). 'Social indifference curves,' *Quarterly Journal of Economics*, vol. 70 (1), pp. 1-22.

- Schultz, T. Paul (1990). 'Testing the neoclassical model of family labor supply and fertility,' *Journal of Human Resources*, vol. 25(4), pp. 599-634.
- Thomas, Duncan (1990). 'Intra-household resource allocation: an inferential approach,' *Journal of Human Resources*, vol. 25(4), pp. 635-664.
- U.K. Central Statistical Office (1984-1987). Annual Abstract of Statistics, London: HMSO.
- U.K. Department of Employment and Productivity (1973-1976, 1980-1983). *Family Expenditure Survey*, London: HMSO.
- U.K. Department of Health and Social Security (1991). *Social Security Statistics*, London: HMSO.
- U.K. House of Commons Hansard, January 1980.
- Wales, T.J. and Woodland, A.D. (1983). 'Estimation of consumer demand systems with binding non-negativity constraints,' *Journal of Econometrics*, vol. 21(3), pp. 263-285.

Table 1: Summary Statistics

| Variable Broadly-defined goods Mean' Std Dev Mean' Std Dev observations Broadly-defined goods Alcohol 0.043 (0.047) 2.041 (2.745) 13474 Clothing 0.078 (0.076) 4.059 (5.205) 14953 Durables 0.063 (0.091) 3.761 (9.932) 15559 Food in 0.231 (0.087) 9.976 (3.759) 16131 Food out 0.034 (0.032) 1.693 (2.035) 15014 Food out 0.034 (0.032) 1.693 (2.035) 15014 Fuel, power, & light 0.062 (0.041) 2.701 (1.991) 16085 Housing 0.150 (0.079) 7.016 (7.173 16109 Misc 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.036 (0.043) 1.491 (1.675) 10426 <t< th=""><th></th><th>Share of</th><th></th><th>Real I</th><th></th><th># positive</th></t<> | | Share of | | Real I | | # positive |
|---|------------------------|--|----------|--|----------|--------------|
| Alcohol 0.043 (0.047) 2.041 (2.745) 13474 (Clothing 0.078 (0.076) 4.059 (5.205) 14953 Durables 0.063 (0.091) 3.761 (9.932) 15559 Food in 0.231 (0.087) 9.976 (3.759) 16131 Food out 0.034 (0.032) 1.693 (2.035) 15014 Fuel, power, & light 0.062 (0.041) 2.701 (1.991) 16085 Housing 0.150 (0.079) 7.016 (7.173) 16109 Misc 0.085 (0.056) 4.152 (4.458) 16130 Sevices 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.036 (0.043) 1.491 (1.675) 10426 Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.009) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Real total expenditure 48.802 (28.086) 8.078 (0.048) 1.370 (1.659) 9878 Real total expenditure 48.802 (28.086) 2.0391 (1.239) 5987 Real total expenditure less clothing & 3.782 (0.436) Predicted log real expenditure 48.802 (28.086) 8.078 (0.069) (0.099) 0.199 (0.483) 5987 Real total expenditure less clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | Variable | $\mathbf{Mean}^{\scriptscriptstyle \dagger}$ | Std Dev | $\mathbf{Mean}^{\scriptscriptstyle \dagger}$ | Std Dev | observations |
| Clothing | · | | | | | |
| Durables 0.063 (0.091) 3.761 (9.932) 15559 Food in 0.231 (0.087) 9.976 (3.759) 16131 Food out 0.034 (0.032) 1.693 (2.035) 15014 Fuel, power, & light 0.062 (0.041) 2.701 (1.991) 16085 Housing 0.150 (0.079) 7.016 (7.173) 16109 Misc 0.085 (0.056) 4.152 (4.458) 16130 Sevices 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.036 (0.043) 1.491 (1.675) 10426 Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 | | | | | , , | |
| Food in 0.231 (0.087) 9.976 (3.759) 16131 Food out 0.034 (0.032) 1.693 (2.035) 15014 Fuel, power, & light 0.062 (0.041) 2.701 (1.991) 16085 Housing 0.150 (0.079) 7.016 (7.173) 16109 Misc 0.085 (0.056) 4.152 (4.458) 16130 Sevices 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.036 (0.043) 1.491 (1.675) 10426 Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.027 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.041) 1.008 (2.695) 5906 | Clothing | | ` , | | | |
| Food out Food out Fuel, power, & light Fuel, power, on the light Fuel, power, on the light Fuel, power, & light Fuel, power, on the | | | ` , | | , , | |
| Fuel, power, & light 0.062 (0.041) 2.701 (1.991) 16085 Housing 0.150 (0.079) 7.016 (7.173) 16109 Misc 0.085 (0.056) 4.152 (4.458) 16130 Sevices 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.036 (0.043) 1.491 (1.675) 10426 Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) <td>Food in</td> <td>0.231</td> <td>` ,</td> <td></td> <td>` ,</td> <td></td> | Food in | 0.231 | ` , | | ` , | |
| Housing 0.150 | Food out | | | | ` , | |
| Misc 0.085 (0.056) 4.152 (4.458) 16130 Sevices 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.036 (0.043) 1.491 (1.675) 10426 Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 < | Fuel, power, & light | | (0.041) | 2.701 | (1.991) | |
| Sevices 0.086 (0.082) 4.798 (9.520) 16113 Tobacco 0.036 (0.043) 1.491 (1.675) 10426 Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 < | Housing | 0.150 | (0.079) | 7.016 | | 16109 |
| Tobacco 0.036 (0.043) 1.491 (1.675) 10426 Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.009) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 | Misc | 0.085 | (0.056) | 4.152 | (4.458) | 16130 |
| Transport 0.132 (0.111) 6.949 (11.530) 15815 Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 < | Sevices | 0.086 | (0.082) | 4.798 | (9.520) | 16113 |
| Narrowly-defined goods Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.04 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) | Tobacco | 0.036 | (0.043) | 1.491 | (1.675) | 10426 |
| Women's clothing 0.027 (0.042) 1.445 (2.740) 11046 Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) 20.262 20.262 20.262 | Transport | 0.132 | (0.111) | 6.949 | (11.530) | 15815 |
| Men's clothing 0.018 (0.041) 1.008 (2.695) 5906 Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) Log real total expenditure 3.782 (0.262) Real total expenditure less clothing & durables 20.469 (18.205) <td>Narrowly-defined goods</td> <td></td> <td></td> <td></td> <td></td> <td></td> | Narrowly-defined goods | | | | | |
| Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 3.782 (0.262) Real total expenditure less clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | Women's clothing | 0.027 | (0.042) | 1.445 | (2.740) | 11046 |
| Kid's clothing 0.027 (0.039) 1.327 (2.176) 10469 Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 3.782 (0.262) Real total expenditure less clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | Men's clothing | 0.018 | (0.041) | 1.008 | (2.695) | 5906 |
| Cigarettes 0.034 (0.043) 1.370 (1.659) 9429 Other tobacco 0.003 (0.010) 0.121 (0.434) 2457 Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) 20.436) 20.436) 20.436) 20.436) 20.469 20. | Kid's clothing | 0.027 | | 1.327 | (2.176) | 10469 |
| Domestic services 0.004 (0.015) (0.019) (0.899) 2362 Cosmetics 0.005 (0.009) (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) (0.010) (0.256 (0.519) 5805 Books 0.003 (0.010) (0.010) (0.185 (0.694)) 5676 Toys 0.008 (0.022) (0.391 (1.239)) 5954 Pets expenditure 0.008 (0.018) (0.363 (0.978)) 8167 Children's pocket money 0.004 (0.009) (0.483) 5987 Real total expenditure 48.802 (28.086) 20.436) Log real total exp 3.782 (0.436) Predicted log real expenditure 3.782 (0.262) Real total expenditure less clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | <u> </u> | 0.034 | , , | 1.370 | , , | 9429 |
| Domestic services 0.004 (0.015) 0.199 (0.899) 2362 Cosmetics 0.005 (0.009) 0.251 (0.509) 9816 Hairdressing 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) 20.436) 20.436) 20.436) 20.469 18.205) Real total expenditure less clothing & durables 20.469 (18.205) 20.469 18.205) Real normal income (from earnings) 46.767 (37.864) 37.864) 37.864 | Other tobacco | 0.003 | (0.010) | 0.121 | (0.434) | 2457 |
| Hairdressing Books 0.005 (0.010) 0.256 (0.519) 5805 Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) Log real total exp 3.782 (0.436) Predicted log real expenditure 3.782 (0.262) Real total expenditure less clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | Domestic services | 0.004 | (0.015) | 0.199 | (0.899) | 2362 |
| Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) | Cosmetics | 0.005 | (0.009) | 0.251 | (0.509) | 9816 |
| Books 0.003 (0.010) 0.185 (0.694) 5676 Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) | Hairdressing | 0.005 | (0.010) | 0.256 | (0.519) | 5805 |
| Toys 0.008 (0.022) 0.391 (1.239) 5954 Pets expenditure 0.008 (0.018) 0.363 (0.978) 8167 Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) | _ | 0.003 | ` , | 0.185 | ` , | 5676 |
| Pets expenditure 0.008 (0.018) (0.018) (0.978) 8167 Children's pocket money 0.004 (0.009) (0.009) (0.483) 5987 Real total expenditure 48.802 (28.086) (28.086) Log real total exp 3.782 (0.436) Predicted log real expenditure 3.782 (0.262) Real total expenditure less clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | Toys | 0.008 | ` , | 0.391 | ` , | 5954 |
| Children's pocket money 0.004 (0.009) 0.199 (0.483) 5987 Real total expenditure 48.802 (28.086) Log real total exp 3.782 (0.436) Predicted log real expenditure 3.782 (0.262) Real total expenditure less clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | • | 0.008 | , , | 0.363 | , , | 8167 |
| Log real total exp Predicted log real expenditure Real total expenditure less clothing & durables (from earnings) 3.782 (0.436) (0.262) (18.205) (18.205) | - | | ` , | | | |
| Log real total exp Predicted log real expenditure Real total expenditure less clothing & durables (from earnings) 3.782 (0.436) (0.262) (18.205) (18.205) | | | | | | |
| Predicted log real expenditure Real total expenditure less clothing & durables Real normal income (from earnings) 3.782 (0.262) 20.469 (18.205) 46.767 (37.864) | <u> </u> | | • • | | | |
| Real total expenditure less clothing & durables Real normal income (from earnings) 20.469 (18.205) 46.767 (37.864) | • | | • • | | | |
| clothing & durables 20.469 (18.205) Real normal income (from earnings) 46.767 (37.864) | | 3.782 | (0.262) | | | |
| Real normal income (from earnings) 46.767 (37.864) | | | | | | |
| (from earnings) 46.767 (37.864) | • | 20.469 | (18.205) | | | |
| | | | /a= a= | | | |
| N = 16131 | (from earnings) | 46.767 | (37.864) | | | |
| | N = 16131 | | | | | |
| # children in family Frequency Percent | # children in family | Frequency | Percent | | | |
| 1 5513 34.2 | <u> </u> | | | | | |
| 2 7721 47.9 | 2 | | | | | |
| 3 2897 18.0 | | 2897 | | | | |

[†] Means and standard deviations are for full sample.

Expenditure levels are £ per week in 1974 currency.

Mean shares are computed using total expenditure. Means of shares computed using expenditure excluding clothing and durables are proportionally larger, and are not constrained to be less than or equal to unity.

Table 2: Instrumented Broadly-Defined Goods (a) Budget Shares, TSLS

| | Durables | Food in | Fuel, Power | Housing | Misc. | Services | Transport |
|----------------|-----------|------------|------------------------------|--------------|-----------|-----------|-----------|
| Policy dummy | -0.0033 | -0.0112** | 0.0004 | -0.0314*** | 0.0087*** | 90000 | 0.0038 |
| 1 child family | (0.0058) | (0.0045) | (0.0026) | (0.0058) | (0.0032) | (0.0045) | (0.0061) |
| Policy dummy | -0.0041 | -0.0022 | -0.0004 | -0.0353*** | 0.0074** | 0.0018 | -0.0035 |
| 2-child family | (0.0055) | (0.0043) | (0.0026) | (0.0058) | (0.003) | (0.0043) | (0.0058) |
| Policy dummy | 0.0007 | -0.0056 | 0.0021 | -0.0354*** | 0.0082** | 0.0031 | -0.001 |
| 3-child family | (0.0061) | (0.0048) | (0.0027) | (0.006) | (0.0034) | (0.0051) | (0.0065) |
| Log real | 0.0299*** | -0.1349*** | -0.0375*** | -0.004 | ***8900.0 | 0.0793*** | 0.0617*** |
| expenditure | (0.0032) | (0.0024) | (0.0014) | (0.0028) | (0.0019) | (0.0032) | (0.0038) |
| Log own real | 0.0728 | | 0.036*** | 0.1298*** | 0.1358*** | 0.0499 | 0.0657 |
| price | (0.0444) | | (0.0112) | (0.0235) | (0.0318) | (0.0311) | (0.0631) |
| \mathbb{R}^2 | 0.0738 | 0.4279 | 0.1916 | 0.1056 | 0.0511 | 0.0689 | 0.0610 |
| F statistic | 0.68 | 5.62*** | 0.85 | 13.08*** | 2.64** | 0.20 | 1.17 |
| | | | (b) Expenditure Levels, TSLS | Levels, TSLS | | | |
| | Durables | Food in | Fuel, Power | Housing | Misc. | Services | Transport |
| Policy dummy | -0.115 | -0.293 | -0.02 | -1.825*** | 0.453* | 0.125 | 0.276 |
| 1 child family | (0.586) | (0.218) | (0.131) | (0.466) | (0.26) | (0.48) | (0.419) |
| Policy dummy | 0.023 | 0.018 | -0.054 | -2.103*** | 0.205 | 0.294 | -0.229 |
| 2-child family | (0.623) | (0.208) | (0.133) | (0.505) | (0.217) | (0.479) | (0.415) |
| Policy dummy | 0.333 | -0.187 | 0.084 | -1.971*** | 0.396* | 0.544 | 0.455 |
| 3-child family | (0.666) | (0.243) | (0.14) | (0.537) | (0.235) | (0.601) | (0.84) |
| Log real | 6.265*** | 3.553*** | 1.167*** | 8.035*** | 4.925*** | 10.914** | 10.87*** |
| expenditure | (0.468) | (0.146) | (0.074) | (0.334) | (0.195) | (0.435) | (0.829) |
| Log own real | -0.775 | 4.254** | 1.753*** | 6.157*** | 9.129*** | 4.7 | 4.03 |
| price | (4.74) | (1.913) | (0.552) | (1.846) | (2.388) | (3.17) | (6.815) |
| \mathbb{R}^2 | 0.1654 | 0.2751 | 0.0904 | 0.1779 | 0.2288 | 0.2097 | 0.1735 |
| F statistic | 0.36 | 2.89** | 98.0 | 5.87*** | 1.55 | 0.42 | 1.09 |

Table 2: Instrumented Broadly-Defined Goods (continued) (c) Budget Shares, TSLS and Tobit

| | Alcohol | hol | Tobacco | 03 | Clothing | gu | Food out | = |
|----------------------------------|----------|-----------|------------|-----------|-----------|-----------|-----------|-----------|
| | LSLS | Tobit | TSLS | Tobit | LSLS | Tobit | LSLS | Tobit |
| Policy dummy | -0.0026 | -0.0032 | -0.0014 | -0.0016 | 0.0089** | 0.0108** | 0.0121*** | 0.0137*** |
| 1 child | (.0028) | (0.0033) | (0.0028) | (0.0041) | (0.0044) | (0.0047) | (0.0023) | (0.0024) |
| Policy dummy | -0.0014 | -0.0014 | 0.0001 | -0.0009 | 0.0116*** | * | 0.0104*** | 0.012*** |
| 2 children | (0.0026) | (0.0031) | (0.0027) | (0.0039) | (0.0042) | | (0.0022) | (0.0023) |
| Policy dummy | -0.005* | -0.0052 | 0.0015 | 0.0005 | 0.0111** | | 0.0047* | 0.0061** |
| 3 children | (0.003) | (0.0035) | (0.0029) | (0.0043) | (0.0047) | (0.0052) | (0.0025) | (0.0026) |
| Log real | 0.0035** | 0.0102*** | -0.0445*** | -0.059*** | 0.0248*** | 0.0275*** | 0.0158*** | 0.0181*** |
| expenditure | (0.0015) | (0.0017) | (0.0014) | (0.002) | (0.0025) | (0.0025) | (0.0011) | (0.001) |
| Log own real | -0.0165 | -0.0154 | 0.0005 | 0.0034 | -0.042** | -0.0548** | 0.0183 | 0.0185 |
| price | (0.0213) | (0.0262) | (0.0098) | (0.0143) | (0.0206) | (0.0227) | (0.0164) | (0.0168) |
| R ² or Log likelihood | 0.0335 | 17847.058 | 0.0897 | 10417.399 | 0.0618 | 15414.191 | 0.0166 | 28990.827 |
| F or χ^2 statistic | 1.46 | 3.44 | 98.0 | 0.62 | 2.69** | 10.98** | 15.72*** | 50.87*** |

(d) Expenditure Levels, TSLS and Tobit

| | Alcohol | hol | Tobacco | 033 | Clothing | gu | Food out | ut |
|----------------------------------|----------|----------|-------------|------------|----------|------------|----------|------------|
| | TSLS | Tobit | LSLS | Tobit | TSLS | Tobit | LSLS | Tobit |
| Policy dummy | -0.093 | -0.061 | -0.033 | -0.035 | 0.412 | 0.734** | 0.48*** | 0.605*** |
| 1 child | (0.147) | (0.187) | (0.112) | (0.165) | (0.256) | (0.315) | (0.127) | (0.149) |
| Policy dummy | -0.06 | 0.029 | 0.007 | -0.008 | 0.429* | 0.847*** | 0.379*** | 0.517*** |
| 2 children | (0.131) | (0.178) | (0.109) | (0.159) | (0.251) | (0.302) | (0.122) | (0.145) |
| Policy dummy | -0.277* | -0.159 | 0.065 | 0.081 | 0.447 | 0.981*** | 0.062 | 0.212 |
| 3 children | (0.161) | (0.203) | (0.12) | (0.176) | (0.297) | (0.343) | (0.141) | (0.158) |
| Log real | 2.169*** | * | -0.553*** | -1.03*** | | 5.348*** | 2.445*** | 2.566*** |
| expenditure | (0.143) | (0.099) | (0.057) | (0.079) | | (0.169) | (0.102) | (0.064) |
| Log own real | -1.378 | | 0.264 | 0.189 | -2.057 | -4.084** | 1.272 | 1.620 |
| price | (1.177) | (1.499) | (0.402) | (0.583) | (1.277) | (1.507) | (0.883) | (1.037) |
| R ² or Log likelihood | 0.1035 | -36426.9 | 0.0078 | -28313.502 | 0.2634 | -47207.215 | 0.1767 | -32713.462 |
| F or χ^2 statistic | 1.47 | 2.07 | 0.53 | 1.08 | 1.05 | 9.29** | 9.74*** | 30.39*** |

SUR cross-equation joint hypothesis test that all policy variables=0: $\chi^2(33) = 135.43***$ Estimates include control variables for household composition category, region, and a quadratic trend. All except clothing also include quarter of the year. Instruments for log real expenditure consist of a 5-order polynomial in "normal" income and 3rd order polynomial in regional unemployment rate. In tobit models, predicted log real expenditure based on these regressors is used in estimates.

***, **, * $p \le .01, .05, .10$ respectively (standard errors in parentheses)

F or χ^2 statistic is for the joint hypothesis that all three policy variables have no effect.

Table 3: Instrumented Narrowly-Defined Goods (a) Budget Shares, TSLS and Tobit

| | , | ; | | | , | • | (| • | ; | • |
|----------------------------------|------------|------------|----------|-----------|----------------|------------|------------|------------|----------|----------------|
| | Ciga | Cigarettes | - | Lopacco | Domestic | : Services | Cosmetics | netics | Haird | ressing |
| | TSLS | Tobit | LSLS | Tobit | LSLS | Tobit | LSLS | | LSLS | Tobit |
| Policy dummy | -0.0008 | -0.0007 | -0.0006 | -0.0042 | 0.0002 0.0088* | *8800.0 | -0.0016*** | -0.0032*** | -0.0004 | -0.0015 |
| 1 child family | (0.0028) | (0.0044) | (0.0000) | (0.0036) | (0.0008) | (0.0045) | (0.0005) | (0.0007) | (0.0000) | 0006) (0.0015) |
| Policy dummy | 0.0013 | 0.0005 | -0.0012* | -0.0061* | -0.0017** | -0.0036 | **60000- | -0.0023*** | 0.00002 | -0.0007 |
| 2-child family | (0.0027) | (0.0043) | (0.0000) | (0.0034) | (0.0008) | (0.0043) | (0.0005) | (0.0007) | (0.0000) | (0.0014) |
| Policy dummy | 0.0019 | 0.0011 | -0.0005 | -0.0022 | -0.0002 | 0.003 | *6000.0- | -0.0021*** | -0.0006 | -0.0023 |
| 3-child family | (0.0029) | (0.0047) | (0.0000) | (0.0038) | (0.0000) | (0.005) | (0.0005) | (0.0008) | (90000) | (0.0016) |
| Log real | -0.0451*** | -0.0672*** | *9000.0 | 0.0081*** | 0.0094*** | 0.057*** | 0.002*** | 0.0046*** | 0.0034** | 0.0125*** |
| expenditure | (0.0014) | (0.0022) | (0.0003) | (0.0017) | (0.0006) | (0.0025) | (0.0003) | (0.0004) | (0.0004) | (0.0008) |
| Log own real | -0.0008 | -0.0035 | 0.0012 | -0.0014 | *7600.0 | 0.0335 | 0.0109** | 0.0183*** | 0.0016 | -0.0032 |
| price | (0.0098) | (0.0157) | (0.0022) | (0.0126) | (0.0058) | | (0.005) | (0.0071) | (0.0041) | (0.0106) |
| R ² or Log likelihood | 0.0825 | 7905.0429 | 0.0026 | 623.23855 | | | 0.0284 | 24958.199 | 0.0043 | 8686.9832 |
| F or χ^2 statistic | 96.0 | 0.43 | 1.97 | 5.05 | 5.07*** | | 3.11** | 18.61*** | 1.04 | 3.23 |

(a) Expenditure Levels, TSLS and Tobit

| | Ciga | Cigarettes | Other | Fobacco Domestic Services | Domesti | Cosn | Cosmetics | Haird | ressing |
|----------------------------------|-----------|------------|----------|---------------------------|----------|---------|------------|---------|-------------------|
| | TSLS | Tobit | LSLS | Tobit | TSLS | TSLS | Tobit | TSLS | Tobit |
| Policy dummy | 900.0 | 0.018 | -0.039 | -0.211 | -0.007 | -0.057* | -0.14*** | -0.042 | -0.093 |
| 1 child family | (0.111) | (0.180) | (0.027) | (0.162) | (0.049) | (0.032) | (0.043) | (0.029) | (0.073) |
| Policy dummy | 0.079 | 0.076 | -0.072** | -0.307** | -0.105** | -0.031 | **660.0- | -0.014 | -0.034 |
| 2-child family | (0.108) | (0.173) | (0.028) | (0.155) | (0.046) | (0.026) | (0.041) | (0.028) | (690.0) |
| Policy dummy | 0.095 | 0.120 | -0.031 | -0.112 | -0.014 | -0.043 | -0.104** | -0.05 | -0.118 |
| 3-child family | (0.12) | (0.191) | (0.029) | (0.173) | (0.058) | (0.029) | (0.047) | (0.032) | (0.032) (0.079) |
| Log real | -0.688*** | -1.417*** | 0.135*** | 0.589*** | 0.86*** | 0.35*** | 0.538*** | 0.44*** | ***926.0 |
| expenditure | (0.056) | (0.086) | (0.017) | (0.076) | (0.058) | (0.025) | (0.024) | (0.024) | (0.040) |
| Log own real | 0.235 | -0.030 | 0.029 | -0.166 | 0.837** | 0.478 | 1.004** | 0.013 | -0.409 |
| price | (0.4) | (0.634) | (0.099) | (0.570) | (0.354) | (0.314) | (0.410) | (0.199) | (0.522) |
| R ² or Log likelihood | 0.0018 | -27098.489 | 6600.0 | -8750.5812 | 0.019 | 0.0887 | -14656.428 | 0.0572 | -13845.98 |
| F or χ^2 statistic | 0.78 | 0.91 | 3.18** | 6.28* | 4.71*** | 1.2 | 10.66** | 1.70 | 4.01 |

Table 3: Instrumented Narrowly-Defined Goods (continued) (c) Budget Shares, TSLS and Tobit

| | Bo | Books | Ь | Pets | Ţ | Foys | Kids' poc | ket money |
|---|----------|-----------|----------|------------|----------|-----------|-----------|---------------------|
| | TSLS | Tobit | TSLS | Tobit | TSLS | Tobit | TSLS | Tobit |
| Policy dummy | 0.0008 | 0.0026* | 0.0011 | 0.0015 | 0.0025** | | 0.0014** | 0.0038** |
| 1 child family | (0.0000) | (0.0014) | (0.0013) | (0.0021) | (0.0011) | (0.0027) | (0.0000) | (0.0006) (0.0016) |
| Policy dummy | 0.0008 | 0.003** | | 0.0012 | 0.0019* | | 0.000 | 0.0028* |
| 2-child family (0.0005) (0.0014) | (0.0005) | (0.0014) | | (0.002) | (0.0011) | | (0.0000) | (0.0015) |
| Policy dummy | 9000.0 | 0.0027* | | 0.0016 | 0.0014 | | 0.0005 | 0.0026 |
| 3-child family | (0.0000) | (0.0016) | | (0.0022) | (0.0013) | | (0.0007) | (0.0016) |
| Log real | 0.004*** | 0.0139*** | | -0.0043*** | -0.0002 | | -0.0003 | 0.0023*** |
| expenditure | (0.0005) | (0.0008) | | (0.001) | (0.0000) | (0.0015) | (0.0003) | (0.0007) |
| Log own real | -0.0115* | -0.0217 | | -0.0322* | 0.0278** | 0.0342 | | |
| price | (0.0064) | (0.0135) | (0.0097) | (0.0182) | (0.0141) | (0.0251) | | |
| R ² or Log likelihood | 0.0072 | 8957.3136 | 0.0136 | 12732.881 | 0.1078 | 6148.9288 | 0.1095 | 11552.691 |
| F or χ^2 statistic | 0.79 | 4.97 | 0.55 | 0.58 | 1.82 | 7.43* | 3.45** | 5.73 |

(d) Expenditure Levels, TSLS and Tobit

| | Bo | Books | 1 | Pets | Ī | Toys | Kids' po | cket money |
|----------------------------------|----------|------------|-------------|------------|-------------|------------|----------|-------------------|
| | TSLS | Tobit | TSLS | Tobit | TSLS | Tobit | TSLS | Tobit |
| Policy dummy | 0.032 | 0.151* | 0.054 | 0.091 | 0.131* | 0.31** | 0.054* | 0.174** |
| >> | (0.04) | (0.092) | (0.064) | (0.115) | (0.069) | (0.154) | (0.03) | (0.081) |
| Policy dummy | 0.033 | 0.189** | 0.065 | 0.079 | 0.107* | 0.175 | 0.027 | 0.123 |
| 2-child family | (0.036) | (0.087) | (0.061) | (0.111) | (0.062) | (0.146) | (0.03) | (0.077) |
| Policy dummy | 0.026 | 0.172* | 860.0 | 0.133 | 960.0 | 0.071 | 0.002 | 0.1111 |
| 3-child family | (0.044) | (0.100) | (0.071) | (0.122) | (0.081) | (0.167) | (0.036) | (0.036) (0.082) |
| Log real | 0.457*** | 1.188*** | 0.27*** | 0.278*** | 0.289*** | 0.795*** | 0.17*** | 0.38*** |
| liture | (0.044) | (0.050) | (0.044) | (0.053) | (0.038) | (0.085) | (0.017) | (0.033) |
| | -0.737* | -1.283 | -0.545 | -1.466 | 1.334 | 1.854 | | |
| | (0.399) | (0.863) | (0.495) | (0.991) | | (1.420) | | |
| R ² or Log likelihood | 0.0320 | -14412.118 | 0.0282 | -19714.104 | | -17730.041 | 0.1217 | -11800.539 |
| F or χ^2 statistic | 0.31 | 4.79 | 0.70 | 1.27 | | 6.11 | 3.1 | 5.16 |

Estimates include control variables for household composition category, region, a quadratic time trend, and quarter of the year.

Instruments for log real expenditure consist of a 5-order polynomial in "normal" income and 3rd order polynomial in regional unemployment rate. In tobit models, predicted log real expenditure based on these regressors is used in estimates.

***, **, * $p \le .01$, .05, .10 respectively (standard errors in parentheses) F or χ^2 statistic is for the joint hypothesis that all three policy variables have no effect.

Table 4: Narrow Clothing, Shares and Expenditure Levels, Instrumented and Proxied Total Expenditure

| | | Men's Clothing | <u> </u> | > | Women's Clothing | ing | <u></u> | Children's Clothing | ing |
|--------------------|-------------|----------------|------------|-----------|------------------|------------|------------|---------------------|------------|
| | Share | Level | Share | Share | Level | Share | Share | Level | Share |
| | LSLS | TSLS | Proxy OLS | LSLS | TSLS | Proxy OLS | LSLS | TSLS | Proxy OLS |
| Policy dummy | -0.0037 | -0.136 | -0.0015 | 0.0002 | 0.016 | 0.003 | 0.0078*** | 0.331** | 0.0123*** |
| 1 child family | (0.0026) | (0.162) | (0.0044) | (0.0025) | (0.144) | (0.0036) | (0.0026) | (0.137) | (0.0036) |
| Policy dummy | -0.004 | -0.183 | -0.001 | 0.0036 | 0.093 | 0.0078** | 0.0063** | 0.292** | 0.0104*** |
| 2-child family | (0.0025) | (0.153) | (0.0042) | (0.0023) | (0.137) | (0.0034) | (0.0026) | (0.141) | (0.0038) |
| Policy dummy | -0.0054** | -0.325* | -0.0048 | 0.0037 | 0.163 | 0.0076** | 0.007** | 0.371** | 0.011*** |
| 3-child family | (0.0027) | (0.176) | (0.0045) | (0.0026) | (0.165) | (0.0037) | (0.003) | (0.162) | (0.0041) |
| Log real total | 0.0078*** | 1.468*** | -0.0141*** | 0.0175*** | 2.349*** | -0.0132*** | -0.001 | 1.091*** | -0.0105*** |
| expenditure | (0.0014) | (0.116) | (0.0012) | (0.0014) | (0.124) | (0.001) | (0.0013) | (0.09) | (0.001) |
| Log own real | 0.0129 | 0.657 | 0.0217 | -0.0196** | -1.216** | -0.0249* | -0.0266** | -0.982 | -0.0299* |
| price | (0.0127) | (0.791) | (0.0203) | (9600.0) | (0.604) | (0.0142) | (0.0122) | (0.647) | (0.0169) |
| Trend | 0.0068 | 0.3 | 0.02 | 9600.0 | *969.0 | 0.0147 | -0.0316*** | -1.671*** | -0.0357*** |
| | (0.0078) | (0.497) | (0.0141) | (6900:0) | (0.413) | (0.0112) | (0.0093) | (0.504) | (0.0138) |
| Trend ² | -0.00004 | -0.002 | -0.0001 | -0.0001* | -0.005* | -0.0001 | 0.0002*** | 0.01*** | 0.0002** |
| | (0.0001) | (0.003) | (0.0001) | (0.00005) | (0.003) | (0.0001) | (0.0001) | (0.003) | (0.0001) |
| constant | -0.2854 | -16.577 | -0.7007 | -0.3264 | -29.905* | -0.3665 | 1.3423*** | 65.67*** | 1.5692*** |
| | (0.2954) | (18.807) | (0.5297) | (0.2664) | (15.808) | (0.4233) | (0.375) | (20.251) | (0.5437) |
| ${f R}^2$ | 0.0281 | 0.0932 | 0.1008 | 0.0527 | 0.1556 | 0.1061 | 0.0342 | 0.1159 | 0.1049 |
| F statistic | 1.35 | 1.37 | 0.89 | 2.18* | 0.61 | 2.97** | 3.07** | 2.17* | 3.91*** |

Instruments for log real expenditure in TSLS and two-stage tobit estimates consist of a 5-order polynomial in "normal income" and 3rd order polynomial in N=16131 for women's and men's clothing; N=14073 for children's clothing due to exclusion of 1973 for which there is no children's clothing price index. regional unemployment rate. In proxy OLS estimates, log real total expenditure excludes expenditures on clothing and durables. ***, **, * $p \le .01$, .05, .10 respectively (standard errors in parentheses)
F statistic is for the joint hypothesis that all three policy variables have no effect All estimates also include control variables for household composition category, and region.

Appendix A

Table A1: OLS Estimate of Log Real Total Expenditure Using Instruments as Regressors

| O.000703 | Normal income | 0.011637*** |
|---|------------------------------|------------------------|
| (| rmal income ² | (0.000709) 3.29F-06 |
| ((() () () () () () () () () | | (9.90E-06) |
| (| rmal income ³ | -2.11E-07*** |
| (rly | | (5.40E-08) |
| | rmal income ⁴ | 5.53E-10*** |
| rly (rate² (rate³ (rate³ (| | (1.17E-10) |
| cent rate (rate² (rate³ - | rmal income ⁵ | -4.12E-13*** |
| rate (| | (8.41E-14) |
| ((| gional quarterly | -0.03113*** |
| | unemployment rate | (0.006575) |
| | employment rate ² | 0.003034*** |
| - | • | (0.000808) |
| | employment rate ³ | -9.4E-05*** |
|) | | (2.91E-05) |
| (0.022718) | nstant | 3.19473*** |
| | | (0.022718) |

Standard errors in parentheses. *** $p \le .01$

0.3613

 $Adj. R^2$

Appendix B: Broad and Narrow Goods Shares Using Proxy for Total Expenditure

Table B1: Broadly-Defined Goods, Budget Shares, Proxy for Total Expenditure
(a) OLS

| | Durables | Food in | Fuel, Power | Housing | Misc. | | Transport |
|----------------|------------|------------|-------------|------------|------------|-----------|-----------|
| Policy dummy | 0.024*** | -0.0294*** | -0.0071** | -0.0274** | *4900.0 | 0.0147*** | 0.0041 |
| 1 child family | (0.0084) | (0.0062) | (0.0033) | (0.0064) | (0.0039) | | (0.0071) |
| Policy dummy | 0.0243*** | -0.0227*** | ***600.0- | -0.0304*** | 0.0062* | | -0.0012 |
| 2-child family | (0.0086) | (0.006) | (0.0032) | (0.0063) | (0.0036) | | (0.0068) |
| Policy dummy | 0.023** | -0.0224*** | -0.0052 | -0.0306*** | 0.0074* | | 0.0012 |
| 3-child family | (0.009) | (0.0067) | (0.0035) | (0.0067) | (0.0041) | | (0.0076) |
| Log real | -0.1413*** | -0.0119*** | -0.0015*** | -0.0042*** | -0.0026*** | | 0.013*** |
| expenditure | (0.0038) | (0.001) | (0.0003) | (0.0005) | (0.0004) | | (0.001) |
| Own real price | -0.0965 | 0.1286** | 0.0372*** | 0.2209*** | 0.2108*** | | 0.1833** |
| | (0.0599) | (0.0532) | (0.0138) | (0.0269) | (0.0383) | | (0.0722) |
| Adj. R² | 0.7483 | 0.1088 | 0.0654 | 0.0915 | 0.0658 | | 0.0321 |
| F statistic | 2.87** | 7.68*** | 3.54** | 7.98*** | 1.22 | | 0.49 |

(b) OLS and Tobit

| | AI | Alcohol | Tol | opacco | Clo | Jothing | F00 | Food out |
|---------------------------|-----------|-----------|-----------|-----------|------------|------------|------------|------------|
| | OLS | Tobit | OLS | Tobit | OLS | Tobit | OLS | Tobit |
| Policy dummy | -0.0053 | -0.0071* | -0.0072** | -0.0079 | 0.018*** | 0.0179** | 0.028*** | 0.0307*** |
| 1 child | (0.0034) | (0.004) | (0.0034) | (0.0051) | (0.0067) | (0.007) | (0.0029) | (0.0031) |
| Policy dummy | -0.0039 | -0.005 | -0.0066** | -0.0079 | 0.0225*** | 0.0233*** | 0.0265*** | 0.029*** |
| 2 children | (0.0032) | (0.0038) | (0.0033) | (0.0049) | (0.0068) | (0.0067) | (0.0028) | (0.003) |
| Policy dummy | -0.0074** | **600.0- | -0.0036 | -0.0029 | 0.0189*** | 0.0197*** | 0.0195*** | 0.0212*** |
| 3 children | (0.0036) | (0.0043) | (0.0036) | (0.0054) | (0.0072) | (0.0075) | (0.0031) | (0.0032) |
| Log real | 0.0003 | *20000 | 0.0001 | **6000.0 | -0.0398*** | -0.0411*** | -0.0011*** | -0.0011*** |
| expenditure | (0.0003) | (0.0004) | (0.0002) | (0.0004) | (0.0028) | (90000) | (0.0002) | (0.0002) |
| Own real price | -0.0395 | -0.0468 | 0.0019 | -0.0023 | -0.0391 | -0.041 | -0.0227 | -0.0293 |
| ı | (0.0249) | (0.0306) | (0.0115) | (0.0171) | (0.0307) | (0.0322) | (0.0197) | (0.0203) |
| Adj. R² or Log likelihood | 0.0356 | 15959.503 | 0.0348 | 8690.7554 | 0.2354 | 10750.979 | 0.0419 | 26544.058 |
| F or χ^2 statistic | 1.68 | 5.37 | 2.29* | 5.03 | 3.67** | 12.43*** | 37.97*** | 121.56*** |

Estimates include control variables for household composition category, region, and a quadratic trend. All except clothing also include quarter of the year.

***, **, * $p \le .01, .05, .10$ respectively (standard errors in parentheses)

Log real expenditure, not including clothing and durables expenditures, is used as a proxy for log real total expenditure. F or χ^2 statistic is for the joint hypothesis that all three policy variables have no effect

Table B2: Narrowly-Defined Goods, Budget Shares, Proxy for Total Expenditure

| | Ciga | Cigarettes | Other | Other Tobacco | Domestic | Domestic Services | Cosmetics | netics | Hairdı | Hairdressing |
|-------------------------|----------|------------|-----------|---------------|-----------|-------------------|---------------|------------|------------|--------------|
| | OLS | Tobit | OLS | Tobit | OLS | Tobit | | Tobit | OLS | Tobit |
| Policy dummy | -0.0062* | -0.0063 | -0.001 | -0.0061 | 0.0026** | 0.0156*** | -0.0017** | -0.0032*** | -0.0003 | -0.002 |
| 1 child family | (0.0034) | (0.0056) | (0.0007) | (0.0044) | (0.0011) | (0.0056) | $\overline{}$ | (0.0009) | (0.0007) | (0.0019) |
| Policy dummy | -0.0051 | -0.0059 | -0.0015** | *800.0- | 0.0007 | 0.0019 | -0.0006 | -0.0019** | 0.0004 | -0.0007 |
| 2-child family | (0.0033) | (0.0054) | (0.0008) | (0.0043) | (0.001) | (0.0054) | (90000) | (0.0009) | (0.0007) | (0.0018) |
| Policy dummy | -0.0029 | -0.0014 | -0.0007 | -0.0039 | 0.0021* | 0.0076 | -0.0007 | -0.0021** | -0.0004 | -0.0031 |
| 3-child family | (0.0036) | (0.0059) | (0.0008) | (0.0047) | (0.0011) | (0.0061) | (90000) | (0.001) | (0.0008) | (0.002) |
| Log real | 0.0003 | 0.0014*** | -0.0002** | -0.0003 | 0.0004*** | 0.0034*** | -0.0004*** | -0.0004*** | -0.0004*** | ***9000.0- |
| expenditure | (0.0002) | (0.0005) | (0.0001) | (0.0003) | (0.0001) | (0.0000) | (0.0001) | (0.0001) | (0.0001) | (0.0002) |
| Log own real | 0.0000 | -0.0077 | 0.001 | -0.0043 | 0.0103 | 0.0392 | 0.0152** | 0.0232*** | -0.0028 | -0.0138 |
| price | (0.0114) | (0.0188) | (0.0026) | (0.0148) | (0.0068) | (0.0387) | (0.0063) | (0.0087) | (0.0049) | (0.0129) |
| Adj. R ² or | | | | | | | | | | |
| Log likelihood | 0.0340 | 6255.3166 | 0.0058 | 245.3711 | 0.0181 | -490.78157 | 0.0331 | 23067.56 | 0.0182 | 7565.9824 |
| F or χ^2 statistic | 1.64 | 3.12 | 1.95 | 5.01 | 4.36*** | 18.94*** | 2.78** | 12.72*** | 1.21 | 4.39 |

| | В | ooks | | Pets | T | Poys | Kids' poc | ket money |
|--|----------|-----------|----------|----------|-----------|-----------|--------------------|-----------|
| | OLS | Tobit | OLS | Tobit | | Tobit | OTS | Tobit |
| Policy dummy | 0.0015** | 0.0042** | 0.0007 | 0.0005 | 0.0015 | 0.0051 | 0.0013* | 0.0026 |
| 1 child family | (0.0007) | (0.0018) | (0.0014) | (0.0025) | (0.0014) | (0.0034) | (0.0007) | (0.002) |
| Policy dummy | 0.0016** | 0.0051*** | 0.001 | 0.0003 | 6000.0 | 0.0022 | 9000.0 | 0.0014 |
| 2-child family | (0.0007) | (0.0017) | (0.0013) | (0.0024) | (0.0014) | (0.0033) | (0.0008) | (0.0019) |
| Policy dummy | 0.0013* | 0.004** | 0.0014 | 0.001 | 0.0003 | -0.0011 | 0.0001 | 0.0011 |
| 3-child family | (0.0007) | (0.0019) | (0.0016) | (0.0026) | (0.0016) | (0.0037) | (0.0009) (0.002) | (0.002) |
| Log real | 80000000 | 0.0003** | -0.0001 | 0.0002 | -0.0002* | -0.0003 | -0.0002*** | -0.0002 |
| expenditure | (0.0001) | (0.0002) | (0.0002) | (0.0002) | (0.0001) | (0.0003) | (0.0001) | (0.0001) |
| Log own real | -0.0111 | -0.0213 | -0.015 | -0.0339 | 0.0466*** | 0.0592* | | |
| price | (0.0075) | (0.0162) | (0.0108) | (0.021) | (0.0176) | (0.0306) | | |
| Adj. R ² or Log likelihood 0.0125 | 0.0125 | 7904.1474 | 0.0102 | 11596.07 | .01115 | 5089.7417 | 0.1135 | 10662.361 |
| F or χ^2 statistic | 1.97 | 9.12** | 0.36 | 0.27 | 99.0 | 6.55* | 3.30** | 2.92 |

SUR cross-equation joint hypothesis test that all policy variables=0: χ^2 (33) = 275.94*** Estimates include control variables for household composition category, region, a quadratic time trend, and quarter of the year. Log real expenditure, not including clothing and durables expenditures, is used as a proxy for log real total expenditure. ***, ** p \leq .01, .05, .10 respectively (standard errors in parentheses) For χ^2 statistic is for the joint hypothesis that all three policy variables have no effect